Author's response to reviews

Title: Cost-effectiveness analysis of pemetrexed versus docetaxel in the second-line treatment of non-small cell lung cancer in Spain: results for the non-squamous histology population

Authors:

Yumi Asukai (yasukai@uk.imshealth.com)
Amparo Valladares (valladares_maria_amparo@lilly.com)
Carlos Camps (camps_car@gva.es)
Eifiona Wood (ewood@uk.imshealth.com)
Kaisa Taipale (taipale_kaisa@lilly.com)
Jorge Arellano (arellano_jorge@lilly.com)
Alejo Cassinello (cassinello_alejo@lilly.com)
José Antonio Sacristán (sacristan_jose@lilly.com)
Tatiana Dilla (dilla_tatiana@lilly.com)

Version: 4 Date: 14 January 2010

Author's response to reviews: see over
Melissa Norton  
Editor-in-Chief, BMC Series  
BioMed Central Ltd, Floor 6  
236 Gray’s Inn Road  
London WC1X 8HL  
United Kingdom

14 January 2010

Dear Ms Norton

I write on behalf of our research team to present our revised manuscript in response to the request for final formatting prior to publication. We are delighted to have our manuscript accepted for publication, and have addressed all major and minor revisions in this re-submission. A list of revisions made is included below for clarity.

Major revisions
- Abstract has been resubmitted and matches what is contained in the manuscript.
- All figures (Figures 1-6) are cited within the manuscript

Minor revisions
- They city is provided in all four affiliation details (Please note the city is Windlesham, not Surrey for the Eli Lilly UK location)
- All table and figure placement texts have been removed from the main text.
- All track changes have been removed from the manuscript.
- Blank page 28 has been removed.
- Reference List has been renamed ‘References’ (Please note automatic re-generation of reference by Reference Manager may cause it to revert to Reference List)
- Tables have been re-formatted to erase visible vertical lines, but invisible vertical lines to define cells are still in tact.
- Merged cells crossing rows have been split and text placed in the appropriate cell in lieu.
- Tables 4 and 7 have had their repeating headers removed.
- All figures have been cropped and resaved as JPEG files to minimise white space.

Thank you very much for your response, and please let me know if any other questions should arise.

Yours Sincerely,

Yumi Asukai

On behalf of:

M. Amparo Valladares  
Carlos Camps  
Efiona Wood  
Kaisa Taipale  
Jorge Arellano  
Alejo Cassinello  
José Antonio Sacristán  
Tatiana Dilla